Author’s response to reviews

Title: Community Mobilization and Maternal Care of Women Living With HIV in poor settings: The case of Mfuwe, Zambia

Authors:

Choolwe Muzyamba (muzyamba@merit.unu.edu)

Wim Groot (w.groot@maastrichtuniversity.nl)

Sonila Tomini (tomini@merit.unu.edu)

Milena Pavlova (m.pavlova@maastrichtuniversity.nl)

Version: 1 Date: 14 Oct 2016

Author’s response to reviews:

Dear Sir/Madam,

Thank you very much for your response to our paper. Based on your recommendation, we have removed the waiver letter and also removed from the manuscript the identifying pieces of information (such as age and marital status).

Kind regards,

Choolwe Muzyamba

(Corresponding author)